Author's response to reviews

Title: A Rare Cause of Small Bowel Obstruction: Phytobezoar in a Jejunal Diverticulum.

Authors:

mohammad tayeb (drmtayeb@yahoo.com)
faiz khan (faizkhan.surg@yahoo.com)
fozia rauf (farauf@yahoo.com)
mumtaz khan (mmutazkhan@gmail.com)

Version: 2 Date: 4 January 2011

Author's response to reviews:

1: Abstract written in the formate of introduction, case presentation and conclusion.
2: Ethnicity of patient given
3: Competing interests section filled
4: Authors contribution section filled
5: Written informed consent from patient available for publication of this manuscript and images